Author's response to reviews

Title: Longitudinal proxy measurements in multiple sclerosis: patient-proxy agreement on the impact of MS on daily life over a period of two years.

Authors:

Femke FAH van der Linden (fah.vdlinden@vumc.nl)
Jolijn JJ Kragt (jj.kragt@vumc.nl)
Margarethe M van Bon (msk700@student.vu.nl)
Martin M Klein (m.klein@vumc.nl)
Alan AJ Thompson (a.thompson@ion.ucl.ac.uk)
Henk HM van der Ploeg (vanderploeg-stapert@planet.nl)
Chris CH Polman (ch.polman@vumc.nl)
Bernard BMJ Uitdehaag (bmj.uitdehaag@vumc.nl)

Version: 4 Date: 13 February 2008

Author's response to reviews: see over
Dear Sir,

Please find attached a revised version of our manuscript entitled:
‘Longitudinal proxy measurements in multiple sclerosis: patient-proxy agreement on the impact of MS on daily life over a period of two years’ (MS: 1430757510156619)

The following changes were made:

- Title: the full stop at the end of the title was removed.
- Equal contributions: only two authors are now listed as equal contributors.
- The subheading on page 11 and page 15 were changed into "Factors affecting...."
- On page 16 ‘an improvement OF agreement’ was changed into ‘an improvement IN agreement’
- On page 17 an extra interspace in line 8 between ‘time’ and ‘[19,37]’ was removed.

With kind regards,

Yours sincerely,

Ms. F.A.H van der Linden